Author's response to reviews

Title: N-nitroso-N-ethylurea Activates DNA Damage Surveillance Pathways and Induces Transformation in Mammalian Cells

Authors:

Satish Bodakuntla (satishshiva.dos@gmail.com)
Libi Anandi V (libianandi1211@gmail.com)
Surojit Sural (surojitsural@gmail.com)
Prasad Trivedi (Prasad.trivedi@gmail.com)
Mayurika Lahiri (mayurika.lahiri@iiserpune.ac.in)

Version: 4 Date: 9 April 2014

Author's response to reviews: see over
April 09, 2014

Ref: MS: 2071180488113553

Title: N-nitroso-N-ethylurea Activates DNA Damage Surveillance Pathways and Induces Transformation in Mammalian Cells

Authors: Satish Bodakuntla, Libi Anandi V, Surojit Sural, Prasad Trivedi and Mayurika Lahiri

The Editor
BMC Cancer
BioMed Central Journals

We have made the necessary changes as suggested by the Handling Editor as mentioned below:

1. Please could you provide figure legends for your supplementary/additional files after your current figure legends.

The figure legends for the additional files have been provided below the current figure legends in the revised version.

2. Please could you also provide email addresses for all authors on your title page.

The email addresses of all the authors have been provided on the title page.

I sincerely apologise for over-looking the above-mentioned instructions to authors.

Best regards

[Signature]

Mayurika Lahiri